Author's response to reviews

Title: Protein-bound polysaccharide from Phellinus linteus inhibits tumor growth, invasion, and angiogenesis and alters Wnt/beta-catenin in SW480 human colon cancer cells

Authors:

Kyoung-Sub Song (limlab1@gmail.com)
Ge Li (ybhospital@hanmail.net)
Jong-Seok Kim (livingstream@nate.com)
Kaipeng Jing (limlab4@gmail.com)
Tae-Dong Kim (Tae-Dong-Kim@ouhsc.edu)
Jin-Pyo Kim (jinpkim@hanmail.net)
Seung-Bo Seo (winnerhifive@hanmail.com)
Jae-Kuk Yoo (yjk9191@empal.com)
Hae-Duck Park (dsoonmd@kyuh.co.kr)
Byung-Doo Hwang (bdhwang@cnu.ac.kr)
Kyu Lim (kyulim@cnu.ac.kr)
Wan-Hee Yoon (whyoon@cnu.ac.kr)

Version: 4 Date: 22 July 2011

Author's response to reviews: see over
Dear Editor,

I would like to thank you for your letter regarding our manuscript (Manuscript ID 8581952545269282). Based on your request, we have made some changes on the original manuscript. Here, we attached the proofread manuscript in the format of MS word for your approval. In addition, all the changes we made in this manuscript are listed below.

**Major revisions (we require the author to make these changes)**

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Please write affiliations 1 and 2 in full.

**Reply:** We further detailed the affiliations of authors as requested (title page, lines 5-10 from top).

Once again, thanks for taking your time to edit this manuscript.

Sincerely yours,
Kyu Lim, Ph.D.
Professor

**Corresponding author:** Kyu Lim, Department of Biochemistry, College of Medicine, Chungnam National University, Moonhwa-dong, Joong-gu, Daejeon, 301-747, Korea. Tel: 82-42-580-8223; Fax: 82-42-580-8121; E-mail: kyulim@cnu.ac.kr